Author's response to reviews

Title: What doctors think about the impact of managed care tools on quality of care, costs, autonomy, and relations with patients

Authors:

Marie Deom (mariedeom@gmail.com)
Thomas Agoritsas (thomas.agoritsas@hcuge.ch)
Patrick A Bovier (patrick.bovier@post.harvard.edu)
Thomas V Perneger (thomas.perneger@hcuge.ch)

Version: 4 Date: 6 December 2010

Author's response to reviews:

This revision responds to editorial/production requests.

Appendix 1 was removed; the content was included in the main text (top half of page 6).

Appendix 2 was removed, as it overlaps with the content of Table 2. We have added the following at the bottom of page 8: (Table 2; full item distributions of the items are available from the last author on request).

All minor changes were made as well.

With best regards

Thomas Perneger